

NOTA DI LAVORO 64.2015

Integrated Assessment of Climate Catastrophes with Endogenous Uncertainty: Does the Risk of Ice Sheet Collapse Justify Precautionary Mitigation?

Delavane B. Diaz, Department of Management Science and Engineering, Stanford University, Stanford, USA

Climate Change and Sustainable Development Series Editor: Carlo Carraro

Integrated Assessment of Climate Catastrophes with Endogenous Uncertainty: Does the Risk of Ice Sheet Collapse Justify Precautionary Mitigation?

By Delavane B. Diaz, Department of Management Science and Engineering, Stanford University, Stanford, USA

Summary

Greenhouse gas policies confront the trade-off between the costs of reducing emissions and the benefits of avoided climate change. The risk of uncertain and potentially irreversible catastrophes is an important issue related to the latter, and one that has not yet been well incorporated into economic models for climate change policy analysis. This paper demonstrates a multistage stochastic programming framework for catastrophe modeling with endogenous uncertainty, applied to a benchmark integrated assessment model. This study moves beyond recent catastrophe or tipping point studies with arbitrary risk, instead investigating the specific threat of the uncertain collapse of the West Antarctic Ice Sheet (WAIS), characterized in accordance with recent expert elicitations, empirical results, and physical relationships. The stochastic DICE-WAIS model introduced here informs risk management strategies that balance uncertain future climate change impacts with the costs of mitigation investments today. This work finds that accounting for the consequences of the possible WAIS collapse in a stochastic setting with endogenous uncertainty leads to more stringent climate policy recommendations (increasing the CO2 control rate by an additional 4% of global emissions and raising the social cost of carbon by \$10), reflecting the need to hedge against uncertainties with downside risk as well as pursue precautionary mitigation.

Keywords: Climate Change Policy, Sea Level Rise, Ice Sheet Collapse, Endogenous Uncertainty, Stochastic Optimization, Greenhouse Gas Mitigation, Risk Management

JEL Classification: C61, D61, D81, H23, Q54, Q58

Originally presented at European Summer School in Resource and Environmental Economics San Servolo, Venice, Italy, July 12, 2014

Address for correspondence Delavane B. Diaz Department of Management Science and Engineering Stanford University Stanford CA 94305 USA Phone: 650-646-5448 E-mail: delavane@stanford.edu Web: profiles.stanford.edu/delavane-diaz

Integrated Assessment of Climate Catastrophes with Endogenous Uncertainty: Does the Risk of Ice Sheet Collapse Justify Precautionary Mitigation?

Delavane B. Diaz[†]

FEEM Note di Lavoro/Working Paper Series originally presented at European Summer School in Resource and Environmental Economics San Servolo, Venice, Italy July 12, 2014

Abstract

Greenhouse gas policies confront the trade-off between the costs of reducing emissions and the benefits of avoided climate change. The risk of uncertain and potentially irreversible catastrophes is an important issue related to the latter, and one that has not yet been well incorporated into economic models for climate change policy analysis. This paper demonstrates a multistage stochastic programming framework for catastrophe modeling with endogenous uncertainty, applied to a benchmark integrated assessment model. This study moves beyond recent catastrophe or tipping point studies with arbitrary risk, instead investigating the specific threat of the uncertain collapse of the West Antarctic Ice Sheet (WAIS), characterized in accordance with recent expert elicitations, empirical results, and physical relationships. The stochastic DICE-WAIS model introduced here informs risk management strategies that balance uncertain future climate change impacts with the costs of mitigation investments today. This work finds that accounting for the consequences of the possible WAIS collapse in a stochastic setting with endogenous uncertainty leads to more stringent climate policy recommendations (increasing the CO₂ control rate by an additional 4% of global emissions and raising the social cost of carbon by \$10), reflecting the need to hedge against uncertainties with downside risk as well as pursue precautionary mitigation.

Keywords: Climate change policy, sea level rise, ice sheet collapse, endogenous uncertainty, stochastic optimization, greenhouse gas mitigation, risk management **JEL:** C61, D61, D81, H23, Q54, Q58

[†]Department of Management Science and Engineering, Stanford University, Stanford, CA 94305; Phone: 650-646-5448; Email: delavane@stanford.edu; Web: profiles.stanford.edu/delavane-diaz.

1 Introduction

Greenhouse gas policies confront the trade-off between the costs of reducing emissions and the benefits of avoided climate change. However, there are unique challenges related to the long time horizon of the problem (with temporal dynamics such as thermal inertia and other lags), the heterogeneous nature of climate impacts, the sizable economic toll of both action and inaction (which occur on different timescales, and are thus contingent on the discount rate), and the pervasive uncertainty around the physical science and the resulting damages.

To help address these multiple dimensions and complexities, researchers in the field of integrated assessment have developed models for interdisciplinary climate and energy policy analysis. Integrated assessment models (IAMs) are a key framework for exploring the implications and consequences of alternative global climate change scenarios (Weyant et al., 1996; Parson and Fisher-Vanden, 1997). These models are widely used and relied upon by policymakers to evaluate climate proposals and inform policy design. In fact, the US government recently used three IAMs to compute the social cost of an incremental carbon emission for use in cost-benefit analysis of federal regulations (IAWG, 2010, 2013).

However, these particular models have faced renewed criticism of their representation of climate damages (Revesz et al., 2014; Pindyck, 2013; Stern, 2013).¹ These damages to society are typically represented in IAMs with a damage function, which in its most basic form equates a future amount of temperature change to an absolute or fractional loss of economic output.² Despite the fact that there is a limited empirical basis for the damage functions, they are often extrapolated to different regions and applied beyond current observations of warming. This deficiency is compounded by the fact that most IAMs are deterministic, meaning that all parameter values are fully known, and many optimize with perfect foresight. These characteristics are incongruent with the cascades of uncertainties that permeate the causal chain of climate change (Schneider, Kuntz-Duriseti, and Azar, 2000).

1.1 Approaches to Modeling Uncertainty

Uncertainty can be characterized as parametric (epistemic), due to imperfect knowledge, or stochastic (aleatory), due to inherent randomness and natural variability (Kann and Weyant, 2000;

¹Early IAM studies led to unresolvable debates over distributional concerns related to the aggregation of impacts by region and time (i.e., the discount rate) as well as the sheer uncertainty about impacts. In response, the costbenefit approach was largely abandoned in favor of the cost-effectiveness approach which sets an environmental goal and reformulates the problem to minimize costs while meeting the target. These controversial topics are now back in the spotlight due to their implications for the social cost of carbon being applied to US regulations (IAWG, 2010, 2013).

²Impact sectors such as agriculture and energy have existing markets that inform damage estimates, whereas the valuation of impacts to intangible goods such as ecosystems and human health are even more difficult.

Dessai and Hulme, 2003). While parametric uncertainty can be reduced as knowledge about physical dynamics and social systems improves over time, the variability that characterizes stochastic uncertainty will always exist to some extent. Rather than incorporate these types of uncertainty directly into the model structure, most IAMs have been developed with a deterministic structure, and uncertain parameters are investigated with ex-post analysis. These approaches are classified as uncertainty propagation methods (Kann and Weyant, 2000), and include scenario and sensitivity analysis of key uncertain parameters (e.g., climate sensitivity, damage coefficients, or the discount rate) that are varied in single- or multi-parameter mode, and the Monte Carlo approach, which samples from a range of distributions.³ However, these uncertainty propagation methods return a collection of possible outcomes, each corresponding to different conditions, without identifying a single strategy that performs well in the face of uncertainty.⁴

In 'Buying Greenhouse Insurance,' Manne and Richels (1991) describe uncertainty propagation models as the 'learn, then act' framework, in contrast to the 'act, then learn' framework, which requires decisions to be made before uncertainty is resolved. The latter is often a more useful and realistic model of decision-making in the real world. 'Act, then learn' frameworks incorporate decision-making under uncertainty directly into the IAM structure, such that a single robust strategy is followed until uncertainty is resolved. This is known as a hedging strategy, because it is designed to perform reasonably well given the range of possible uncertain outcomes.

Often a decision-maker's prior beliefs about uncertainty may change over time. This could take the form of learning about the true value of an unknown parameter in the case of parametric uncertainty, or Bayesian updating of a probability distribution to reflect new evidence in the case of stochastic uncertainty. A framework that recognizes learning over time by a decision-maker is known as sequential decision-making. There are several examples of sequential decision-making applied to certain dimensions of climate change problem, such as technological change, socioeconomic growth, and climate sensitivity (Manne and Richels, 1995; Kelly and Kolstad, 1999; Webster, Jakobovits, and Norton, 2008; Webster, Santen, and Parpas, 2012; Bistline and Weyant, 2013).

Two state-of-the-art stochastic optimization techniques used to model sequential decisionmaking under uncertainty are stochastic programming and stochastic dynamic programming. Stochastic programs are structured such that parametric uncertainty is resolved after one or more stages. If the model allows recourse, the decision-maker can respond to the information learned and compensate accordingly (Birge and Louveaux, 1997). Dynamic programming is a solution algorithm that decomposes a complex multi-stage optimization problem into a series of simpler subproblems, which are then solved with backward recursion (Bertsekas, 1976). Stochastic dy-

³Uncertainty propagation via Monte Carlo simulation is the approach used by two well-known IAMs, PAGE and FUND (Hope, 2011b; Anthoff and Tol, 2013).

⁴There are decision-making frameworks that address the issue of a suite optimal results: a preferred strategy can be selected on the basis of alternative evaluation criteria such as minimax regret or maximin welfare (Bergh, 2004) or robust decision-making (Lempert and Schlesinger, 2000; Lempert, Popper, and Bankes, 2003).

namic programming represents uncertainty with various states that describe the evolution of the system over time. The appropriate method for a particular research question will depend on the type of uncertainty considered (e.g., parametric or stochastic), as well as the dimensionality of the problem.⁵

1.2 Stochastic Optimization with Climate Thresholds

A special type of stochastic uncertainty in the greenhouse gas problem is that of a climate catastrophe, a low-probability, high-consequence event that would lead the earth system to shift abruptly in response to climate change (Weitzman, 2009, 2012). Examples of potential catastrophes are ocean thermohaline circulation (THC) disruption, sudden methane release from the oceans or permafrost, disintegration of the polar ice sheets, and widespread forest die-back (National Research Council, 2002; Alley et al., 2003; Overpeck and Cole, 2006; Keller, Yohe, and Schlesinger, 2007; Lenton et al., 2008). The existence of these threats is supported by the geologic record, but the governing dynamics and trigger thresholds are still not fully understood or quantified due to insufficient data and process models limitations (Kriegler and Hall, 2009).

Nevertheless, the often-cited rationale for stringent mitigation targets is the need for an insurance policy against such climatic catastrophes. Yet there is a disconnect between this policy rhetoric and the supporting analysis: the few IAMs that represent catastrophes tend to use the expected value approach, which eliminates the element of risk, and thus they do not call for aggressive mitigation (Nordhaus and Boyer, 2000; Hope, 2006, 2011a). If society is to purchase such an insurance policy by investing in stringent mitigation, that cost will need to be justified by the magnitude of the catastrophic threat. Thus, there is a real need for IAMs to explore the implications of potential catastrophes with more fidelity in a stochastic setting with sequential decision-making, particularly as the frontier of scientific understanding of catastrophe advances over time.

It is not a trivial task to apply stochastic optimization methods to IAMs, which to begin with have non-linear, dynamic structures with long time horizons and many variables in a deterministic mode. An early, pioneering effort to model uncertain climate thresholds in an IAM with global stochastic optimization was Keller, Bolker, and Bradford (2004). Their study investigated the possible collapse of THC, triggered by passing an uncertain threshold level of carbon dioxide concentrations as described in Stocker and Schmittner (1997). The consequence of the catastrophe was a persistent economic loss ranging from 0 to 3% based on Tol (1998). The paper developed several probabilistic approaches to model this uncertain catastrophe, evolving from sensitivity analysis covering the entire parameter space of both climate sensitivity and economic damage ('learn, then act') to a stochastic program with recourse ('act, then learn'). The problem was formulated with three states

⁵Both methods suffer from the curse of dimensionality, however stochastic programs can accommodate high dimension in the decision variables but only a small number of stages, whereas dynamic programs are better suited for many stages but are limited to a smaller state space of the system variables.

of the world with an exogenously specified probability distribution about climate sensitivity and thus the critical CO_2 threshold, such that the ultimate likelihood of collapse depended on mitigation. They found that parameter uncertainty about the THC collapse threshold led to decreased near term mitigation (e.g., the optimal hedging path is to wait until uncertainty is resolved before investing in mitigation).

While the work of Keller, Bolker, and Bradford (2004) was an important initial step toward incorporating endogenous catastrophes into IAMs, it lacked an endogenous link between the contemporaneous climate and the probability of being in a catastrophic state of the world, instead adopting exogenous parametric uncertainty. In this case the hedging path reflects downside risk aversion. However, a characteristic of climate catastrophes is that their likelihood is thought to increase with warming (Kriegler and Hall, 2009; Urban and Keller, 2010). Stochastic optimization with endogenous uncertainty can overcome this limitation because it allows the probability of catastrophe to be controlled by the climate variables (e.g., CO_2 concentration or temperature). This endogenous approach introduces a precautionary component to the optimal path, not solely hedging against the catastrophic outcome but also altering the probabilities of that state.

Cai, Judd, and Lontzek (2013) and Lemoine and Traeger (2014) are two recent efforts to apply stochastic dynamic programming to model uncertain climate thresholds (also referred to as tipping points) in an IAM.⁶ Both studies use a hazard rate that is endogenously linked to atmospheric temperature to represent the probability of encountering the threshold, which leads to an irreversible regime shift if crossed. Cai, Judd, and Lontzek (2013) (hereafter CJL) consider the possible collapse of THC. If this event occurs there is a transition from the pre-threshold regime, characterized by the baseline damage function, to the post-threshold regime, characterized by a permanent shock to economic output via an upward-shifted damage function. The hazard rate function is calibrated to an expert elicitation of the likelihood of THC collapse temperature change (Zickfeld et al. 2007).

Lemoine and Traeger (2014) (hereafter LT) compare the effect of two potential tipping points, activating just one at a time: the first increases the climate sensitivity and the second prolongs the lifetime of carbon dioxide in the atmosphere. Unlike the economic consequences incurred in the post-tip regime of CJL, crossing the threshold in LT alters the physical response of the climate system. The authors assume a uniform prior distribution for the threshold temperature (e.g., the threshold lies somewhere between current temperature and an arbitrary upper bound with equal likelihood). Over time, as warmer temperatures are observed and tolerated without causing the system to tip, the Bayesian hazard rate is updated by truncating the portion of the probability distribution that has been safely observed.

A major difference between the two designs is how the hazard rate changes over time with

⁶Both of these stochastic dynamic programs are solved with backward recursion by decomposing the problem into the pre-threshold and post-threshold regime and approximating the value functions. Details on the respective algorithms can be found in Cai, Judd, and Lontzek (2012); Traeger (2013).

warming. CJL formulate the hazard rate as a Markovian jump process that is a linear function of current surface temperature. In this way the probability of encountering the tipping point jumps discretely to a new value, to face a higher probability of catastrophe if the climate continues to warm or a lower probability if temperatures have begun to decline. This allows for the fact that a tipping point can occur even after the climate begins to cool (e.g., at a temperature that has already been experienced) because there is still a positive hazard rate. In contrast, LT assume that there is a certain tipping point at some unknown but unique temperature. Accordingly, the Bayesian hazard rate is continually updated, eliminating 'safe' temperatures from the probability distribution and concentrating the weight on the remaining 'risky' temperatures. This formulation guarantees that a tipping point will not occur after the climate has been stabilized or begins to cool. Of these two approaches, the Markovian appears to offer more skill for representing physical system tipping points where risk persists for any forcing over pre-industrial.

A third optimization technique for sequential decision-making under uncertainty is stochastic programming with endogenous uncertainty (Goel and Grossmann, 2006; Dupacova, 2006). This differs from the classic stochastic programming approach, in which the probability distributions of all uncertain parameters are exogenously fixed a priori, because the probability distributions are instead an endogenous function of the decision variables. There is a limited and scattered literature on stochastic programming applications with endogenous uncertainty. It has been applied to the offshore gas field development planning problem (Goel and Grossmann, 2004) and the research and development portfolio management problem (Solak et al., 2010).

The potential for using stochastic programming with endogenous uncertainty to incorporate a generic catastrophe into an IAM has been recently demonstrated by Rutherford (2013), using an alternative name, stochastic control. This particular approach will be discussed in more detail in Section 3. A clear advantage of endogenous uncertainty in stochastic programming is that it overcomes the limitation of traditional stochastic programming to credibly represent the responsive nature of a climate change catastrophe described above. Furthermore, in contrast to dynamic programming with endogenous uncertainty, a stochastic program of this type can often be solved as a single optimization problem. This simpler approach avoids the computational burden of backward recursion and value function approximation required in dynamic programming, and can reduce solution times by more than an order of magnitude.

An important next phase in modeling catastrophes with endogenous hazard in IAMs is to progress beyond the generic, stylized catastrophe. While CJL specify the THC collapse as their particular catastrophe, the resulting consequence is an arbitrary economic penalty, not based on an empirical or economic assessment. LT model abrupt changes that manifest consequences in accordance with the physical system; however, the hazard is characterized with the stylized assumption of a uniform prior, which implies the threshold for an abrupt climatic regime shift is equally likely to happen anywhere between today's temperature and an arbitrary upper bound. While the contributions of these papers are notable and valuable as methodological demonstrations, the characterization of catastrophes can and should be improved if such analysis is to be useful for policy-makers. For each specific catastrophe that threatens society, both the hazard and the consequence should be characterized with fidelity, based on physical relationships, empirical results, or expert elicitation, and reflect unique features such as hysteresis or irreversibility. To the extent that these key characteristics can be represented, the better will be the policy guidance.

1.3 Present Work

This paper demonstrates a stochastic programming framework for modeling catastrophes with endogenous uncertainty in an IAM following the approach of Rutherford (2013). I apply his framework to the uncertain collapse of the West Antarctic Ice Sheet (WAIS). In order to study this particular catastrophe, I modify the stochastic structure and augment the IAM with additional climatic variables so that the ice sheet collapse is characterized in keeping with the uncertain, irreversible nature of the event, as well as the physical system dynamics, accounting for thermal inertia of the oceans, lags in the mitigation signal, and likely hysteresis. This work finds that accounting for catastrophic consequences in a stochastic programming setting with endogenous uncertainty leads to more stringent climate policy recommendations, reflecting the need to hedge against uncertainties with downside risk as well as pursue precautionary mitigation.

The remainder of this paper will be organized as follows. Section 2 provides the relevant scientific background on sea level rise and the role of the cryosphere in particular. Section 3 introduces the stochastic programming framework for a climate catastrophe with endogenous uncertainty, as well as the unique characterization of the risk of WAIS collapse. Section 4 reformulates a benchmark integrated assessment model with the stochastic programming framework and describes modifications to incorporate WAIS climate variables and sea level rise impacts. Section 5 presents results, showing the effect of catastrophe risk on the optimal mitigation policy as well as quantifying the role of precautionary mitigation and the expected value of perfect information. Section 6 discusses policy implications, limitations, and directions for future work.

2 Ice Sheets and Global Sea Level Rise

Climate change will directly affect the two main components of global sea level: thermal expansion and loss of land ice (Milne et al., 2009). Thermal expansion occurs when the oceans absorb additional heat from the atmosphere: water expands as it warms which increases the volume of seawater. Loss of land ice, is generally subdivided into two groups: mountain glaciers and small ice caps, and the massive polar ice sheets of Greenland and Antarctica. Mountain glaciers and small ice caps contain enough water to increase sea levels by 0.32 m. Of greater concern is the possibility of rapid melt from the disintegration of the Greenland and Antarctic ice sheets, which store approximately 65 m of sea level rise (SLR) combined (Vaughan, Comiso, and Allison, 2013). Greenland holds over 7 m of SLR. A mountain range in Antarctica separates the smaller West Antarctic Ice Sheet (WAIS), with nearly 5 m of SLR, from the larger and more stable the East Antarctic Ice Sheet (EAIS), which stores over 53 m.

Ice sheets contribute to SLR when annual snowfall additions are less than mass loss to either incremental surface melt, also called ablation or runoff, or dynamic ice discharge (Bamber, Layberry, and Gogineni, 2001; van den Broeke et al., 2009). The former is driven by warming air temperature. In the case of the Greenland there is a known positive feedback cycle in which ablation lowers the surface elevation, thereby exposing the ice sheet to warmer air temperature at the lower altitude (Vaughan, Comiso, and Allison, 2013). The latter is driven by complex ice dynamics, including changes in ocean temperature and circulation (Bamber and Aspinall, 2013). Ice sheets are most vulnerable to discharge at their outer perimeter, where ice shelves serve as a buttress, or support structure, to slow the flow of ice streams into the ocean (Rignot et al., 2004).

WAIS is particularly at risk of dynamic discharge because it is a marine ice sheet. Much of its base is grounded below sea level with floating ice shelves that are exposed to warming subsurface ocean temperatures that cause basal melt (Joughin and Alley, 2011; Church and Clark, 2013). Moreover, it is potentially subject to the 'marine ice sheet instability' (Weertman, 1974), because it has a reverse bedslope that deepens inland, increasing the force from the ice sheet's own weight pushing outward toward the ocean (Joughin and Alley, 2011). WAIS has experienced major ice shelf collapses in the last decade (Shepherd, Wingham, and Rignot, 2004; Bamber et al., 2009), and two recent studies of WAIS by independent scientists argue that the current extent of loss has destabilized the ice sheet irreversibly (Joughin, Smith, and Medley, 2014; Rignot et al., 2014).

Specialized ice sheet models are used to simulate behavior through either surface mass balance or ice flow dynamics. Early models focused on surface mass balance and often projected that warming would increase net mass due to increased snow fall. Now they focus on representing ice flow at the grounding line, the critical boundary between floating marine ice and grounded ice (Rignot et al., 2014). These efforts are currently thought to be incomplete, facing obstacles such as computational limitations to resolve the migrating calving front or grounding line in three-dimensional space. The bottom line is that despite ongoing research and progress, SLR projections are uncertain and fail to account for abrupt changes from a possible ice sheet collapse, so there may be surprises ahead that are worth preparing for.⁷

Many negative impacts are expected to accompany these increases in sea levels. The scientific

⁷In fact, this has happened in the past: a study of paleo-climatic data found that during the last warm interglacial period (over 100,000 years ago) global average sea level rose at an average rate of 1.5 m per century to a level 5 m above present, when mean temperatures were 2 °C above pre-industrial and the polar ice sheets had melted (Rohling et al., 2008; Church and Clark, 2013).

literature and popular media warn that rising sea levels will affect tens of millions of people who live in low-lying coastal areas, as well as infrastructure and capital assets, vulnerable ecosystems, cultural heritage, and many of the most densely populated and economically productive cities in the world. The potential damages from SLR and storm surge to coasts span a range of impacts including accelerated erosion of beaches and cliffs, permanent inundation of low-lying zones, salinization of water resources, degradation of coastal wetlands and increased flooding events (Dasgupta et al., 2007; Nicholls et al., 2007; Nicholls, 2011; Hinkel et al., 2013). Climate impact assessments consistently find that the coastal sector comprises a significant component of the estimated costs of climate change (Nordhaus and Boyer, 2000; Tol, 2002a,b).

Effective policies for baseline SLR will require risk-management strategies that consider the long-term nature of the threat which will unfold gradually over several centuries due to thermal inertia in oceanic processes such as mixing and circulation. However, the very same inertia that moderates the rate of SLR also ensures that SLR will not be easily reversed even with aggressive greenhouse gas mitigation, a feature known as *commitment* to SLR.⁸ This condition is compounded by the abrupt SLR that would be triggered by an ice sheet collapse.

3 Stochastic Programming with Endogenous Uncertainty

In the next two sections I present a stochastic programming IAM with endogenous uncertainty that investigates the possible collapse of WAIS and the optimal policy response. In this section I explain the multistage stochastic programming framework, based on the approach of Rutherford (2013), and in Section 4 I use the framework to reformulate a simple IAM and describe modifications to incorporate WAIS climate variables and SLR impacts.

3.1 Rutherford's Stochastic Catastrophe Framework

Rutherford (2013) presents a sequential, binomial scenario tree to represent the occurrence of a climate catastrophe in a stochastic programming framework, as depicted in Figure 1 below. This parsimonious description of the catastrophic 'states of the world' allows the optimization problem to be efficiently solved via a deterministic equivalent formulation. The multistage stochastic program imitates the more complex structure of the stochastic dynamic programs employed in Cai, Judd, and Lontzek (2013) and Lemoine and Traeger (2014), but avoids the computational burden of backward recursion and value function approximation, reducing solve times by more than an order of magnitude.

⁸If greenhouse gas concentrations were stabilized by 2100, SLR would continue into the future due to thermal expansion of the oceans; the rate of thermal expansion would decrease only gradually over many centuries (Meehl et al., 2007).



Figure 1: Scenario tree for stochastic programming framework for modeling a climate catastrophe. The hazard rate hr_t gives the probability of catastrophe in period t conditional on there having been no catastrophe up to period t. Modified from Rutherford (2013)

A catastrophe can be triggered in any time period t, where the hazard rate hr_t represents the conditional probability of a catastrophe occurring based on warming between period t and the next period, assuming one has not yet occurred. This example uses 10 year time periods but the framework can be flexibly defined.

This approach is based on the traditional definition of risk as a function of the hazard (i.e., probability) and the consequence: $Risk = hazard \times consequence$. At the outset, the catastrophic threshold is uncertain, but the decision-maker knows that it is linked to warming and has a prior belief about the probability. The decision-maker can influence current and future probabilities of catastrophe (i.e., hazard rates) by mitigating CO₂ to reduce climate warming.

Rutherford classifies his catastrophe framework as stochastic control, to distinguish the fact that the hazard rate probabilities are endogenous to the model, compared to a traditional stochastic program with exogenous probabilities for each period (i.e., fixed from the outset). As described in Section 1.2, this is equivalent to stochastic programming with endogenous uncertainty. I will return to the distinction between exogenous and endogenous uncertainty in stochastic programming in Section 5 as it relates to the precautionary motive for mitigation.

Rutherford adopts the hazard rate formulation of Lemoine and Traeger (2014) (see Section 1.2) and assumes a uniform prior distribution such that the probability of the catastrophe occurring at a unique temperature threshold T is uniformly distributed between the current temperature T_0 and some upper bound \overline{T} , at which point the catastrophe will certainly have occurred. The Bayesian hazard rate implied by this prior is updated in each time period t if no catastrophe has occurred following $hr_t = \frac{T_{t+1}-T_t}{\overline{T}-T_t}$. Rutherford assumes that the exogenous consequence of the catastrophe is a steeper damage function such that climate change becomes much more costly to society. Specifically, the catastrophe is characterized with a cubic damage function of temperature

(in addition to the baseline quadratic damage function).

Rutherford's work demonstrates an elegant and a computationally efficient approach to modeling catastrophes using stochastic programming with endogenous uncertainty. While his framework is a major contribution, the stylized, generic nature of the catastrophe limits the usefulness of the findings for policy insight. The hazard rate assumption of a uniformly distributed threshold oversimplifies the physical dynamics of climatic catastrophes. Similarly, the consequence is characterized with an arbitrary penalty, rather than being calibrated to scientific understanding of a specific catastrophic response.

As raised in Section 1.2, this type of risk management study will provide most insight to policy-makers when it depicts a specific and well-characterized catastrophe. Because Rutherford's flexible and tractable framework allows wide latitude for the catastrophe characterization, it is an excellent point of departure for a more precise investigation of a specific climate threat, which will be presented in the next subsection.

3.2 Adaptation of Stochastic Programming Framework to WAIS

I modify Rutherford's framework to model the specific catastrophe of WAIS collapse, capturing the key characteristics described in Section 2. The term 'collapse' suggests a sudden event when in reality the process will unfold over a period of several hundreds of years, slowly raising sea levels. I use collapse to mean the onset of melt from WAIS, and the point of no return.⁹ The remainder of this section will describe the characterization of both the hazard and the consequence of such an event.

3.2.1 Hazard of WAIS Collapse

In this subsection I explain how the probabilities of WAIS collapse are endogenously determined in a manner consistent with the science of WAIS stability and dynamics as described in Section 2. The collapse is triggered by local ocean warming. Shaffer (2014) finds that melting at the base of the Antarctic marine ice shelf depends on the temperature difference between adjacent subsurface ocean temperature and the ice shelf base, the latter fixed to the freezing temperature of sea water. Thus, I assume that the probability of triggering a collapse is driven by the Antarctic ocean subsurface temperature, given by a quadratic hazard rate

$$hr_t = \min[\beta \left(\Delta T_{O_t}\right)^2, 1] \tag{1}$$

⁹These events may not coincide in time. For instance, the point of no return may be crossed before rapid melt begins; similarly, rapid melt may begin, but there is still a window of opportunity to prevent the total disintegration of the sheet. Moreover, it is not guaranteed that the onset of rapid melt will be immediately detected in the observational record.

where hr_t is the hazard rate for period t, β is the collapse coefficient, and ΔT_{Ot} is Antarctic ocean subsurface temperature above the freezing temperature of sea water. (Further explanation of local climate variables such as ΔT_{Ot} will be presented in Section 4.2.)

The hazard rate formulated in Equation 1 is conceptually similar to the one employed by Cai, Judd, and Lontzek (2013), in that it assumes a Markovian jump process. Unlike the Bayesian hazard rate in Rutherford (2013) and Lemoine and Traeger (2014), there is no 'safe' level of warming after stabilization; the catastrophe is always a threat. This hazard rate formulation is the more appropriate choice for characterizing ice sheet disintegration, as ice sheets are inherently unstable (Mercer, 1978). This reflects the fact that basal melt and the marine ice sheet instability will continue to threaten WAIS even if T_O is stabilized. Specifically, it ensures that there is a positive chance of triggering a collapse for any amount of ΔT_O above the freezing temperature of sea water.

Calibrating such a function as Equation 1 requires some subjective belief about the probability of WAIS collapse for a given climate scenario. This likelihood assessment cannot be based on recent results from Atmosphere-Ocean General Circulation Model or Earth System Model projections because of the limitations discussed in Section 2. There are three published expert surveys of the likelihood of WAIS melt. Most recently, Bamber and Aspinall (2013) elicited expert opinion about the contribution of the three polar ice sheet components to global SLR in 2100 and found a mean melt rate for WAIS of 3 mm/yr (with the 95th percentile melt rate of 11.8 mm/yr). Kriegler and Hall (2009) conducted an expert elicitation on the likelihood of five potential climate tipping points occurring by 2200. For WAIS disintegration they found a mid-range probability of 0.6 from core experts assuming a high temperature scenario (3-5.5°C above pre-industrial in 2100). Finally, a risk estimation study by Vaughan and Spouge (2002) assessed an expert panel about the probability of WAIS collapse by 2200. The result of this study was a probability of 0.05 that the melt rate of WAIS would be 10 mm/yr, with a probability of 0.3 that it would be 2 mm/yr. Overall there is a trend of the likelihoods increasing over time. Moreover, two recent observational and modeling studies suggest that a WAIS collapse is nearly certain at some point in the future, assuming a business-as-usual emissions pathway (Joughin, Smith, and Medley, 2014; Rignot et al., 2014).

I perform the calibration based on the Coupled Model Intercomparison Project Phase 5 (CMIP5) representative concentration pathways (RCPs) (Meinshausen et al., 2011). Specifically, I assume that the highest-emitting radiative forcing scenario, RCP8.5, corresponds to a cumulative probability of 0.5 that the onset of WAIS collapse has been triggered between 2010 and 2100. While this probability is arbitrary, it is in the range of the expert elicitations described above for a high warming scenario.

To calibrate the hazard function of ocean temperature change in Equation 1, I solve a system

of equations for β , the desired hazard rate calibration coefficient:

$$P_s = hr_t \Pi_t \quad \forall s, t : s = t \tag{2}$$

$$hr_t = \min[\beta \left(\Delta T_{O_t}\right)^2, 1] \tag{3}$$

$$\Pi_t = \prod (1 - hr_i) \tag{4}$$

$$\sum_{s \le 2100} P_s = 1 \tag{5}$$

where state s and time period t are defined in 10-year steps between 2010 and 2100, P_s is the probability of being in state s, hr_t is the collapse hazard rate, Π_t is the probability that the collapse has not yet occurred at start of time t, and ΔT_O is the local ocean temperature above freezing given by RCP8.5.¹⁰ Solving this system of 31 equations and 31 unknowns (i.e., P_s , hr_t , Π_t , and β) gives $\beta = 0.008$, which means if the climate were to follow an RCP8.5 trajectory, the probability of a collapse having been triggered would be 16.8% by 2050 and 27.9% by 2070.¹¹

This calibration process yields a logistic cumulative distribution function (CDF) of the collapse probability in terms of ΔT_O that is depicted in Figure 2.



Figure 2: Cumulative distribution function (CDF) of the WAIS collapse probability.

3.2.2 Consequence of WAIS Collapse

The consequence of the WAIS collapse is represented in the stochastic framework as a climate system regime shift. Specifically, when the WAIS collapse has been triggered, there is an irreversible change in the SLR dynamics from the baseline process to an altered state in which there is an additional source of melt from WAIS. The timeframe of complete disintegration of WAIS is on the order of 1,000 years, with expert estimates ranging from 400 to 2400 years (National Research Council,

¹⁰Note that physical constraints ensure $\Delta T_O \ge 0$, as the ocean temperature will always be above freezing.

¹¹It is worth noting that the RCP8.5 trajectory assumes higher emissions that many model baseline assumptions. For example, $\beta = 0.008$ in the DICE model business-as-usual scenario with the mitigation level fixed to 0 implies the probability of a collapse having been triggered is 6.8% by 2050, 16.4% by 2070, and 39.2% by 2100.

2013; Oppenheimer, 1998). In the collapse state of the world, the baseline annual rate of WAIS melt is assumed to be

$$\frac{\mathrm{d}SLE_{WAIS_t}}{\mathrm{d}t} = \frac{5}{\tau} \tag{6}$$

where SLE_{WAIS_t} is the SLR-equivalent melt in m, and τ is the lifetime of WAIS, which I assume to be 500 years. This irreversible design intends to reflect current understanding of ice sheet dynamics (e.g., ice sheet growth occurs over tens of thousands of years through accumulated snowfall, and will not be easily re-formed; dynamic instabilities suggest it is not possible to halt the process of ice flow once underway).

Moreover, the baseline rate is assumed to accelerate over time to represent the likely effect of hysteresis. Shaffer (2014) reports that the WAIS melt rate is found to be 'proportional to the product of ocean flow speed and ocean temperature beneath the ice shelf, both of which increase linearly with ocean warming' as found in Holland et al. (2008). In this way, the baseline rate of melt is assumed to increase linearly with the change in the Antarctic ocean subsurface temperature above today, as depicted in Figure 3.



Figure 3: Annual melt rate of WAIS as a function of Antarctic ocean subsurface temperature.

The direct physical consequence of WAIS collapse is the additional rate of SLR described above.¹² In an IAM framework, this physical consequence of SLR has important implications in terms of social welfare. These costs to society are represented through the coastal damage function of SLR, presented in the following section (see Equation 13).

¹²The physical consequence of WAIS collapse is limited to additional SLR. This constrained scope does not fully capture the effect of WAIS collapse on climate system dynamics, which could alter the local weather patterns as well as the albedo (surface reflectivity). In future work it would be interesting to explore this additional dimension of the climate system.

4 Integrated Assessment Model with Stochastic Framework

I now use the stochastic programming framework with endogenous uncertainty described in Section 3 to reformulate a benchmark integrated assessment model (IAM) of optimal growth, the Dynamic Integrated Climate-Economy (DICE) model, version 2013R (Nordhaus and Sztorc, 2013), with essential modifications to incorporate ice sheet climate variables and SLR impacts.

4.1 Dynamic Integrated Climate-Economy (DICE) model

DICE is a transparent and tractable intertemporal optimization model of economic growth and climate impacts for a single region, the world. DICE solves the optimal Pareto problem, which sets the level of greenhouse gas mitigation such that marginal cost of mitigation is equal to the marginal benefit of avoided climate impacts over the model time path.

DICE chooses the optimal path of consumption (trading off carbon mitigation and capital investment) that maximizes the social welfare objective function.¹³

Welfare is the discounted sum of utility over time, where the isoelastic (i.e., constant relative risk aversion) utility function expresses preferences over per capita consumption:

$$W = \sum_{t \in T} \frac{1}{(1+\rho)^t} \left[l_t \frac{(\frac{C_t}{l_t})^{(1-\eta)} - 1}{1-\eta} \right]$$
(7)

where W is total social welfare, C_t is the level of consumption, l_t is the population, ρ , the pure rate of social time preference, and η is the consumption elasticity parameter. Utility, the second term in the formula, is weighted by the social discount factor, the first term. The parameter ρ reflects intertemporal preferences for comparing utility across different generations.

In DICE utility increases in population and per capita consumption, with diminishing marginal utility from the latter. η , the elasticity of marginal utility of consumption, captures aversion to inequality in per capita consumption levels. These two preference parameters in DICE have been calibrated in accordance with the Ramsey growth equation to observed economic outcomes (e.g., interest rates, rates of return on capital) such that ρ is 1.5 and η is 1.45 (Nordhaus and Sztorc, 2013).

A stylized representation of DICE is shown in Figure 4 below.

¹³DICE is a neoclassical model of optimal economic growth premised on the Ramsey rule $r_t = \rho + \eta g_t$, where r_t is the discount rate, ρ is the pure rate of time preference, η is the marginal utility of consumption, and g_t is the per capita growth rate of consumption (Ramsey 1928).



Figure 4: Stylized diagram of the DICE integrated assessment model.

Economic output is determined by a Cobb-Douglas production function of endogenous capital and exogenous labor, with exogenous Hicks-neutral technological change represented by total factor productivity. This output, if unmitigated, has an associated carbon intensity, resulting in greenhouse gas emissions that warm the atmosphere.

The decision variable for carbon mitigation, μ , equals the fraction of emissions from the businessas-usual emissions projection that are avoided through decarbonization. The cost of mitigation (as a proportion of output) is given by a convex power function of μ in which the marginal cost of mitigation increases more than linearly with μ .

Climate damages Ω act as a claim on output, reducing the amount that can be spent on either welfare-improving consumption today or investment in the future capital stock. DICE uses an aggregate damage function that gives the fraction of economic output lost to temperature during time period t, formulated as a quadratic function of ΔT_t , the equilibrium change in global mean surface temperature above preindustrial. The damage functions will be discussed in more detail in the following section. The DICE-2013R model documentation and GAMS code are described in Nordhaus and Sztorc (2013).

4.2 DICE-WAIS: Augmented IAM with WAIS Collapse

I augment DICE to model SLR, local WAIS climate variables, and coastal damages in order to integrate it with the stochastic programming framework presented in Section 3.2.

The modified model, DICE-WAIS, introduces two additional variables related to the local WAIS climate (which drive WAIS hazard), and four variables related to SLR and associated damages from

coastal impacts (which define WAIS consequence). The ocean temperature threshold that triggers the collapse is an uncertain parameter, but the social planner knows the relationship governing the probability of this threat (Equation 1) as well as the consequence of the event (Equation 6).

The DICE-WAIS model is solved along each one of the collapse branches of the stochastic scenario tree presented in Figure 1. The key difference among the different scenario branches is a shift in the SLR regime. The DICE-WAIS diagram in Figure 5 below depicts the blue 'pre-collapse' pathway that corresponds to the upper branch of the scenario tree, while the red 'post-collapse' pathway corresponds to the offshoot collapse scenarios.



Figure 5: Stylized diagram of the DICE-WAIS model.

The climate module in DICE-2013R computes CO_2 concentrations, radiative forcing, and atmosphere and ocean warming. I introduce two local climate variables for WAIS, the Antarctic air temperature, T_A , and the ocean subsurface temperature adjacent to the Antartica, T_O , which drives the hazard of WAIS collapse.

Antarctic air temperature change at sea level can be derived from global mean temperature change using the southern hemisphere polar amplification factor, α_s . The polar amplification effect describes the fact that warming is more pronounced at high latitudes than at the equator. The magnitude of the effect is commonly represented by applying a polar amplification factor to global mean temperature. I define α_s to be 1.2 in accordance with Holland and Bitz (2003):

$$\Delta T_A = \alpha_s \Delta T \tag{8}$$

The local ocean subsurface temperature, T_O , has been empirically estimated from Antarctic air temperature (reduced to sea level), T_A , in Shaffer (2014). Shaffer uses temperature reconstructions

over the past 240,000 years to relate Antarctic surface air temperatures to the average high-latitude (52-70°S) ocean subsurface (200-800m) temperatures. He derives the equation

$$T_O = 0.00690 T_A^2 + 0.439 T_A + 6.39 \tag{9}$$

I also introduce SLR dynamics that is based on RICE-2010, the first and only vintage of the DICE model to explicitly include an SLR module. RICE-2010 decomposes SLR into contributions from four major processes: thermal expansion, melt from glaciers and small ice caps, Greenland Ice Sheet melt, and WAIS melt (Nordhaus, 2010b). I use the first three components to describe a 'baseline SLR' function of temperature:

$$SLR_{t+1} = (r_{TE} + r_{GSIC} + r_{GIS}) \Delta T_t + SLR_t$$
⁽¹⁰⁾

The fourth component, SLR from WAIS melt, is only introduced in the collapse states of the world. It is represented with WAIS rapid melt regime described in the stochastic programming framework Section 3.2.2:

$$\frac{\mathrm{d}SLE_{WAIS_t}}{\mathrm{d}t} = \frac{5}{\tau} \left(T_{Ot} - T_{Ot_0} \right) \tag{11}$$

There are two types of damages in DICE-WAIS, temperature damages and SLR damages. Both damage functions are based on the DICE-2010 model, which distinguishes two categories of impact sectors, coastal impacts and all other noncoastal impacts.(Nordhaus, 2010a).

Temperature damages are a function of ΔT as given in Equation 12. These damages include all market and nonmarket impacts excluding coastal impacts. For these damages, Ω_{TEMPt} gives the fraction of economic output lost to temperature during time period t, formulated as a quadratic function of ΔT_t , the equilibrium change in global mean surface temperature above preindustrial. The severity of the damage function is specified by the linear coefficient a_0 and quadratic coefficient a_1 :

$$\Omega_{TEMPt} = \alpha_1 \Delta T_t + \alpha_2 \Delta T_t^2 = 0.00008 \Delta T_t + 0.002 \Delta T_t^2$$
(12)

The other type of damages is SLR damages. Ω_{SLRt} gives the fraction of economic output lost to coastal damages during time period t, a function of SLR_t .

$$\Omega_{SLR_t} = \beta_1 SLR_t + \beta_2 SLR_t^2 = 0.00518 SLR_t + 0.00306 SLR_t^2$$
(13)

One limitation of the DICE model is the simple representation of mitigation, which allows the fractional level of emissions controlled to fluctuate freely, with no expansion constraint from period

to period (Keller, McInerney, and Bradford, 2008). This simplification fails to capture real-world inertia that limits the rate of decarbonization due to delayed availability of low-emitting technologies, construction lead times, stranded assets, or other capital turnover factors (Ha-Duong, Grubb, and Hourcade, 1997; Richels and Blanford, 2008). Technology inertia is represented in more sophisticated energy system IAMs, but those are not well-suited for the stochastic catastrophe framework presented here because they are too complex and often ignore climate impacts entirely. The final modification to the DICE model is to impose more realistic limits to the mitigation level μ_t with an expansion constraint, similar to that proposed by Keller, McInerney, and Bradford (2008):

$$\mu_{t+1} \le \mu_t + c \tag{14}$$

where the c is the maximum amount of additional mitigation during a 10-year period. c is calibrated to be 0.2 based on analysis of decarbonization rates found in an inter-comparison of 18 energyeconomy and integrated assessment models in Stanford Energy Modeling Forum Study 27 (Kriegler et al., 2014).

5 Results

5.1 Main Results

Figure 6 illustrates two optimal levels of mitigation, one corresponding to a world that is ignorant of the possibility of WAIS collapse and another that understands the nature of that threat, as described in Sections 3 and 4.



Figure 6: Optimal mitigation pathway.

The risk of ice sheet collapse justifies a more stringent climate policy. Specifically, it is optimal to mitigate an additional 4% of global emissions. This is shown by the increase from the optimal level of mitigation when there is no threat of collapse (gray line) up to the optimal pre-collapse path (black line), in which the collapse has not yet occurred but is still a threat.

This pre-collapse mitigation path result is an important finding for policy-makers because it gives the optimal pre-collapse mitigation, where the policy should be set in order to avoid, or delay, the collapse. The difference in the optimal mitigation policies with and without the WAIS risk can be understood by examining the marginal cost of carbon dioxide emissions, also known as the social cost of carbon (SCC), illustrated in Figure 7.



Figure 7: Marginal cost of carbon dioxide for each collapse scenario.

As in Figure 6, the gray line shows the case with no collapse threat, and the black line shows the case in which the collapse is an active threat but has not been triggered. The red lines correspond to the states of the world in which WAIS collapse has been triggered. In each of these collapse scenarios, the marginal cost of carbon departs from the hedging path, rising in response to the acceleration in SLR, which corresponds to a steeper slope on the quadratic damage function given in Equation 13. Because the marginal cost of CO_2 (equivalently the marginal benefit of avoided CO_2) will equal the marginal cost of mitigation along the optimal policy path, each of these red lines show that the value of mitigation becomes even greater after a collapse. The post-collapse marginal cost illustrates the motivation for hedging against the collapse.

Figure 8 shows the temperature and sea level rise trajectories corresponding to each collapse scenario.



Figure 8: Climate outcomes corresponding to each collapse scenario.

Despite the slightly lower temperatures in 2100 in the collapse cases (left), the corresponding SLR outcome for those states is much greater than the baseline pathway (right). The right panel illustrates that the earlier WAIS collapses, the more SLR that society will face, leading to increased damages in the coastal sector.

I also model an alternative scenario tree, in which all uncertainty about the ice sheet collapse is eliminated. There is only one state of the world and it suffers an ice sheet collapse in 2080 with certainty.



Figure 9: Marginal cost of carbon dioxide for certain collapse in 2080.

This single state is given by the blue line. The initial SCC is lower than along the hedging path. This implies there is some additional early mitigation over the no threat case, and mitigation becomes increasingly more beneficial in the periods approaching the collapse.

5.1.1 Robustness Test

The model presented in Sections 3 and 4 is designed to be a convex problem. However, stochastic optimization with endogenous probabilities is computationally challenging and can result in nonconvexities. In order to test that the combination of an endogenous hazard with endogenous consequence has not inadvertently created this situation, I perform a robustness test to show that the solution is indeed the global optimum. Specifically, I test a wide range of initial conditions for the decision variable for carbon mitigation, μ , exploring the entire solution space. Each of these initial conditions returns the same optimal mitigation path, as seen in Figure 10, confirming the robustness of the result.



Figure 10: Robustness test of decision variable initial conditions.

5.2 Precautionary Motivation for Mitigation

The increased mitigation effort from the case of ignorance about the collapse to the pre-collapse level (Figure 6) is due to two factors. The first is the hedging motive due to the fact that the WAIS collapse introduces downside risk from additional SLR damages, so additional mitigation reduces the consequence of entering the collapse state. The second is the precautionary motive due to the fact that the optimizing agent is able to reduce the probability of collapse through additional mitigation that slows the subsurface warming of the Antarctic ocean. These two factors have been shown analytically by Lemoine and Traeger (2014) to be additively separable. This section investigates the relative contribution of the two motives through a numerical experiment. When the model is formulated with endogenous uncertainty, both motives are present. In contrast, in a stochastic program with exogenous uncertainty there is no opportunity for precaution. These two modes can be compared to separate the influence of the two factors.

At the outset of the stochastic program with endogenous uncertainty, the hazard rate at each timestep is unknown, as it depends on the warming that is allowed to occur by the social planner. After solving the model for the optimal pathway, it is possible to observe what the hazard rate was for each time step. I then reformulate the model as a classic multistage stochastic program with probabilities specified exogenously, corresponding to the hazard rates from the original model. This exogenous setting eliminates the precautionary motive.

Figure 11 compares the two formulations of endogenous and exogenous hazard relative to the ignorant case, in terms of the level of mitigation on the left and the marginal cost of carbon dioxide on the right.



Figure 11: Comparison of endogenous and exogenous hazard formulation with optimal mitigation pathway (left) and marginal cost of carbon dioxide (right).

The optimal path can be decomposed into the hedging motive and the precautionary motive based on the relative location of the exogenous hazard (blue line). The collapse risk motive is given by the increase from the result with no threat of collapse (gray line) up to the exogenous hazard hedging path (blue line). The precautionary motive is then given by the increase from exogenous hazard hedging path (blue line) up to the endogenous hazard path (red line). Both panels illustrate that what matters most for mitigation policy is not the existence of the hazard itself, but the fact that it is endogenous and therefore can be reduced.

5.3 Expected Value of Perfect Information

In this section I present the expected value of perfect information (EVPI) about the likelihood of WAIS collapse. The EVPI is defined as the difference between the expected value of the objective function obtained from the 'learn, then act' solution for all states of the world, compared to the expected value of the 'act, then learn' stochastic problem. In a convex maximization problem the former is higher than the latter, since the decision-maker has perfect information for a given state and can behave optimally for that particular scenario in isolation (e.g., without having to satisfy nonanticipativity constraints for coincident states).

In order to quantify this metric in a meaningful unit, the objective function must be converted from units of utility to consumption in a way that preserves the planner's preferences over risk, consumption elasticity of substitution, and time. This requires computing the certain-equivalent consumption in each period t, according to the DICE utility function defined earlier in Equation 7:

$$U_t = \frac{\left(\frac{C_t}{l_t}\right)^{(1-\eta)} - 1}{1-\eta}$$
(15)

Accordingly, \tilde{C}_t , the certain-equivalent consumption, is given by

$$\tilde{C}_t = (\mathbb{E}[U_t^{\omega}](1-\eta) + 1)^{\frac{1}{(1-\eta)}} l_t$$
(16)

Thus, the expected net present value of certain-equivalent consumption, discounted at the consumption discount rate r, is given by

$$z^* = \sum_{t \in T} \frac{1}{(1+r)^t} \,\tilde{C}_t \tag{17}$$

The expected value of discounted consumption for the DICE-WAIS stochastic model with endogenous uncertainty is $z^* =$ \$2,405 Trillion in discounted certain-equivalent consumption.

This z^* can then be compared to the expected value of discounted consumption given perfect information, $z_{ws} = \mathbb{E}[z^{\omega}] = \sum_{\omega \in \Omega} z^{\omega} p(\omega)$, where z^{ω} corresponds to the resulting net present value of consumption for each $\omega \in \Omega$ and $p(\omega)$ is the corresponding state probability realized in the stochastic formulation. Practically, this is computed by solving the 'learn, then act' version of the model for each collapse branch of the scenario tree independently. The results are summarized in Table 1 below.

ω , state of the world	$p(\omega)$	z^{ω}
collapse in 2020	0.6%	\$2984
collapse in 2030	1.2%	\$2982
collapse in 2040	1.9%	\$2980
collapse in 2050	2.7%	\$2978
collapse in 2060	3.6%	\$2976
collapse in 2070	4.4%	\$2975
collapse in 2080	5.2%	\$2973
collapse in 2090	5.8%	\$2972
collapse in 2100	6.2%	\$2971
collapse in 2110	6.4%	\$2970
collapse in 2120	6.4%	\$2969
collapse in 2130	6.3%	\$2968
collapse in 2140	6%	\$2968
collapse in 2150	5.6%	\$2967
no collapse	37.6%	\$2966

Table 1: Discounted global consumption 2010-2250 (Trillion \$).

The expected value of discounted consumption with perfect information about the WAIS collapse is $z_{ws} =$ \$2,969 Trillion.

Finally, the difference between these two quantities gives the EVPI = $z_{ws} - z^* =$ \$564 Trillion. The EVPI can be interpreted as the price that one would be willing to pay in order to gain access to perfect information. This is nearly twenty percent of the total value. This additional value reflects the benefit of delaying mitigation efforts slightly when the precautionary motive is removed and when the collapse period is known with certainty.

6 Discussion

6.1 Summary and Policy Implications

Integrated assessment models have long played an important role in policy analysis for climate change. Recent model inter-comparison exercises showcase development in many dimensions (e.g., improved representation of energy technologies, agriculture, and land use) (Edenhofer et al., 2014; Kriegler et al., 2014), yet there has been much less progress in the area of decision-making under uncertainty. At the same time, there is a growing demand for climate policy analysis that addresses the need for iterative risk management. As noted earlier, IAMs have attracted recent criticism for focusing on deterministic assessments and omitting catastrophic impacts (Revesz et al., 2014; Pindyck, 2013; Stern, 2013); these critiques threaten the relevance of IAMs in the policy arena.

This paper seeks to address these limitations by applying a stochastic programming framework with endogenous uncertainty to a benchmark IAM in order investigate the specific catastrophe of WAIS collapse. This work finds that accounting for the potential ice sheet collapse leads to more stringent climate policy recommendations, increasing the mitigation effort by an additional 4% of global emissions and raising the social cost of carbon by \$10. This analysis is also able to decompose the relative contribution of the hedging and precautionary motive for mitigation and quantify the value of information about this climate uncertainty.

Moreover, this study moves beyond several recent catastrophe or tipping point studies with arbitrary risk, instead investigating the specific threat of the WAIS collapse, characterizing both the hazard and the consequence in accordance with recent expert elicitations, empirical results, and physical relationships. This additional specificity is necessary to move beyond a conceptual demonstration to numerical results that are relevant for policy insight.

This work highlights the fact that the optimal response is strongly influenced by the nature of the physical and social systems that will be affected (details that are unavoidably absent in an arbitrary catastrophe). To illustrate this point, Figure 12 is a diagnostic plot of the response functions of four climate outcomes, air temperature, sea level (non-WAIS components), deep ocean temperature, and subsurface Antarctic ocean temperature, assuming some fixed level of mitigation over the entire century, ranging from 0 to 100%.



Figure 12: Responsiveness to mitigation effort of physical outcomes in 2100.

The slope of each line measures the responsiveness of each climate outcome. The greatest

amount of responsiveness is in air temperature. However, for concern about damages driven by SLR, in the case of coastal impacts, or ocean temperature, in the case of ice sheet collapse, there is much less responsiveness to mitigation. Physical inertia and lags in the natural system play a key role in climate policy design.

6.2 Limitations and Future Work

One notable limitation underlying this work is the simplified representation of learning about the climate catastrophe. This study makes the optimistic assumption that the planner is able to detect whether a catastrophe has been triggered in the current period. In reality a catastrophic event may have three distinct moments: when the catastrophe is triggered (which may or may not be irreversible), when the consequence begins, and when the catastrophe can be clearly detected. These events will likely not coincide in time. The collapse (as modeled) triggers the onset of an additional 1 cm/yr of sea level rise, which would amount to an additional 10 cm in the first period. By comparison, there is no guarantee that existing observational methods, which include remotely-sensed satellite measurements of surface ice-sheet extent and gravitational changes as well as RSLR measurements from the global tide gauge network, will be able to detect this event. More sophisticated models of learning about climate parameters have been demonstrated in Kelly and Kolstad (1999) and Urban et al. (2014). Integrating such frameworks into the catastrophe model would improve the relevance and realism of the results.

Another limitation of these results is that they are contingent on the standard DICE expected utility model. However, this simplified model conflates preferences for risk and intertemporal consumption in a single parameter. A current approach for overcoming this limitation is to apply Epstein-Zin utility, which separates the treatment of risk aversion and consumption smoothing (Epstein and Zin, 1989). While other studies referenced in the literature review do address Epstein-Zin preferences (Jensen and Traeger, 2012; Rutherford, 2013; Cai, Judd, and Lontzek, 2013; Lemoine and Traeger, 2014), the primary focus of this paper was to improve the characterization of the catastrophe in terms of the physical system and scientific understanding about risk. Future extensions to this study will address these utility function concerns.

An ongoing direction for similar studies of this kind is to continue to improve the characterization of catastrophes, both in terms of the hazard and the consequence, so that findings are relevant to policy-makers. This study has attempted to do so for WAIS, but necessary simplifications limit the usefulness of these results. For example, the chosen probability distribution governing the likelihood of collapse (i.e., the hazard rate) is a key assumption that will benefit from scientific advances in ice sheet modeling and studies of past behavior. Similarly, the representation of the WAIS climate dynamics can be improved to better capture the lagged response of subsurface ocean temperature to local WAIS atmospheric warming. This would affect the ability for mitigation to affect the collapse. In addition to these local WAIS dynamics, the global climate module in DICE is a vast simplification of the actual climate system, and introduces its own limitations to this work (van Vuuren et al., 2009).

Finally, the paper demonstrates an approach that could be replicated for other catastrophes. Topics for future studies include ocean circulation disruption, Greenland ice sheet collapse, and carbon-cycle tipping points such as sudden methane release from the oceans or permafrost or widespread forest die-back. For each specific catastrophe that threatens society, both the hazard and the consequence should be accurately characterized based on physical relationships, empirical results, and expert elicitation. To the extent that these characteristics can be represented in the modeling framework, the better the policy guidance will be. Furthermore, modeling singular tipping points overlooks the fact that triggering one could increase the likelihood of another. For example, melting permafrost could lead to a runaway albedo change feedback cycle. It is doubtful that the climate system will only ever experience a single climate tipping point, and to oversimplify the model is unrealistic and also sometimes introduces a perverse incentive for the system to increase emissions after a tipping point has been triggered. Although this adds considerable computational burden, the IAM community can pursue model developments to accommodate this need.

The results of this stochastic programming IAM with endogenous uncertainty demonstrate the importance of incorporating the risk of catastrophe directly into the planner's objective function in a realistic characterization. This type of study is useful for policymakers, so they can begin to understand potential hedging strategies that balance the costs of mitigation investments today with the downside risk of future climate change, as well as the precautionary motive.

7 Acknowledgments

This work was supported by the John A. and Cynthia Fry Gunn Stanford Graduate Fellowship and the US DOE Integrated Assessment Research Program, Grant No. DE-SC005171. The approach has benefited tremendously from insightful comments from Chris Forest, Gerd Infanger, Klaus Keller, Thomas Rutherford, and the participants of the EAERE-FEEM-VIU European Summer School in Resource and Environmental Economics and the SEEPAC seminar at Stanford. I would like to thank Chris Field, Claude Reichard and John Weyant for their helpful feedback on a draft of this paper. All errors and opinions are mine.

References

- Alley, R B, J Marotzke, William D. Nordhaus, J T Overpeck, D M Peteet, R a Pielke, R T Pierrehumbert, P B Rhines, T F Stocker, L D Talley, and J M Wallace. 2003. "Abrupt climate change." Science (New York, N.Y.) 299 (5615):2005–10.
- Anthoff, David and Richard S. J. Tol. 2013. "The uncertainty about the social cost of carbon: A decomposition analysis using fund." *Climatic Change* 117 (3):515–530.
- Bamber, J. L. and W. P. Aspinall. 2013. "An expert judgement assessment of future sea level rise from the ice sheets." *Nature Climate Change* 3 (4):424–427.
- Bamber, Jonathan L, Russell L Layberry, and S P Gogineni. 2001. "A new ice thickness and bed data set for the Greenland ice sheet 1. Measurement, data reduction, and errors." Journal of Geophysical Research 106 (D24):33733–33773.
- Bamber, Jonathan L, Riccardo E M Riva, Bert L A Vermeersen, and Anne M LeBrocq. 2009. "Reassessment of the potential sea-level rise from a collapse of the West Antarctic Ice Sheet." *Science* 324 (5929):901–903.
- Bergh, Jeroen C J M Van Den. 2004. "Optimal climate policy is a utopia : from quantitative to qualitative cost-benefit analysis." 48:385–393.
- Bertsekas, Dimitri P. 1976. "Dynamic programming and stochastic control." .
- Birge, John R and François Louveaux. 1997. Introduction to Stochastic Programming. Springer.
- Bistline, John E. and John P. Weyant. 2013. "Electric sector investments under technological and policy-related uncertainties: a stochastic programming approach." *Climatic Change* 121 (2):143–160.
- Cai, Yongyang, Kenneth Judd, and Thomas Lontzek. 2012. "DSICE: a dynamic stochastic integrated model of climate and economy." .
- Cai, Yongyang, Kenneth L KL Judd, and Thomas S TS Thomas S TS Lontzek. 2013. "The social cost of stochastic and irreversible climate change." *National Bureau of Economic Research*.
- Church, J. A. and Peter U. Clark. 2013. "Sea Level Change." In *Climate Change 2013: The Physical Science Basis*, chap. 13.
- Dasgupta, Susmita, Benoit Laplante, Craig Meisner, David Wheeler, and Jianping Yan. 2007. The impact of sea level rise on developing countries: a comparative analysis, vol. 4136. World Bank Publications.
- Dessai, Suraje and Mike Hulme. 2003. "Does climate policy need probabilities?"
- Dupacova, J. 2006. "Optimization under exogenous and endogenous uncertainty."
- Edenhofer, Ottmar, Ramón Pichs-Madruga, Youba Sokona, E Farahani, S Kadner, K Seyboth, A Adler, I Baum, S Brunner, P Eickemeier et al. 2014. "Climate change 2014: Mitigation of climate change." Working group III contribution to the fifth assessment report of the Intergovernmental Panel on Climate Change. UK and New York.
- Epstein, Larry G and Stanley E Zin. 1989. "Substitution, Risk Aversion, and the Temporal Behavior of Consumption Growth and Asset Returns: A Theoretical Framework." *Econometrica* 57 (4):937–969.

- Goel, Vikas and IE Grossmann. 2006. "A class of stochastic programs with decision dependent uncertainty." *Mathematical programming* 394:355–394.
- Goel, Vikas and Ignacio E. Grossmann. 2004. "A stochastic programming approach to planning of offshore gas field developments under uncertainty in reserves." Computers & Chemical Engineering 28 (8):1409–1429.
- Ha-Duong, M, MJ Grubb, and JC Hourcade. 1997. "Influence of socioeconomic inertia and uncertainty on optimal CO2-emission abatement." Nature 390 (November 1997).
- Hinkel, Jochen, DP van Vuuren, Robert J. Nicholls, and RJT Klein. 2013. "The effects of adaptation and mitigation on coastal flood impacts during the 21st century. An application of the DIVA and IMAGE models." *Climatic Change* :783–794.
- Holland, M. M. and C. M. Bitz. 2003. "Polar amplification of climate change in coupled models." *Climate Dynamics* 21 (3-4):221–232.
- Hope, Chris W. 2006. "The marginal impact of CO2 from PAGE2002: An integrated assessment model incorporating the IPCC's five reasons for concern." *Integrated assessment* 6 (1):19–56.

———. 2011a. "The PAGE09 Integrated Assessment Model: A Technical Description."

———. 2011b. "The Social Cost of CO2 from the PAGE09 Model."

IAWG, US. 2010. "Technical support document: Social cost of carbon for regulatory impact analysis under executive order 12866."

——. 2013. "Technical support document: Technical update of the social cost of carbon for regulatory impact analysis under executive order 12866."

- Jensen, Svenn and Christian Traeger. 2012. "Growth and Uncertainty in the Integrated Assessment of Climate Change."
- Joughin, Ian and Richard B. Alley. 2011. "Stability of the West Antarctic ice sheet in a warming world." Nature Geoscience 4 (8):506–513.
- Joughin, Ian, Benjamin E Smith, and Brooke Medley. 2014. "Marine ice sheet collapse potentially under way for the Thwaites Glacier Basin, West Antarctica." Science (New York, N.Y.) 344 (6185):735–8.
- Kann, Antje and John P. Weyant. 2000. "Approaches for performing uncertainty analysis in largescale energy/economic policy models." *Environmental Modeling & Assessment* 5:29–46.
- Keller, Klaus, Benjamin M Bolker, and David F Bradford. 2004. "Uncertain climate thresholds and optimal economic growth." Journal of Environmental Economics and Management 48 (1):723– 741.
- Keller, Klaus, David McInerney, and David F. Bradford. 2008. "Carbon dioxide sequestration: how much and when?" *Climatic Change* 88 (3-4):267–291.
- Keller, Klaus, Gary W. Yohe, and Michael Schlesinger. 2007. "Managing the risks of climate thresholds: uncertainties and information needs." *Climatic Change* 91 (1-2):5–10.
- Kelly, David L. and Charles D Kolstad. 1999. "Bayesian learning, growth, and pollution." *Journal* of Economic Dynamics and Control 23 (4):491–518.
- Kriegler, Elmar and JW Hall. 2009. "Imprecise probability assessment of tipping points in the climate system." *Proceedings of the National Academy of Sciences* (October 2005).

- Kriegler, Elmar, John P. Weyant, Geoffrey J. Blanford, Volker Krey, Leon E. Clarke, Jae Edmonds, Allen Fawcett, Gunnar Luderer, Keywan Riahi, Richard Richels, Steven K. Rose, Massimo Tavoni, and Detlef P. van Vuuren. 2014. "The role of technology for achieving climate policy objectives: overview of the EMF 27 study on global technology and climate policy strategies." *Climatic Change* 123 (3-4):353–367.
- Lemoine, Derek and Christian Traeger. 2014. "Watch Your Step: Optimal Policy in a Tipping Climate." American Economic Journal: Economic Policy 6 (1):137–166.
- Lempert, RJ, SW Popper, and SC Bankes. 2003. Shaping the next one hundred years: new methods for quantitative, long-term policy analysis. RAND.
- Lempert, RJ and ME Schlesinger. 2000. "Robust strategies for abating climate change." Climatic Change :387–401.
- Lenton, Timothy M, Hermann Held, Elmar Kriegler, Jim W Hall, Wolfgang Lucht, Stefan Rahmstorf, and Hans Joachim Schellnhuber. 2008. "Tipping elements in the Earth's climate system." *Proceedings of the National Academy of Sciences of the United States of America* 105 (6):1786–93.
- Manne, Alan S. and Richard G. Richels. 1991. "Buying greenhouse insurance." *Energy Policy* 19 (6):543–552.
- _____. 1995. "The greenhouse debate: Econonmic efficiency, burden sharing and hedging strategies." The Energy Journal :1–14.
- Meehl, G.A., T.F. Stocker, W.D. Collins, P. Friedlingstein, A.T. Gaye, J.M. Gregory, R. A. Kitoh, J.M. Knutti, A. Murphy, S.C.B. Noda, I.G. Raper, A.J. Weaver Watterson, and Z.-C. Zhao. 2007. "Global Climate Projections." In *Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change*, edited by M. Tignor Solomon, S., D. Qin, M. Manning, Z. Chen, M. Marquis, K.B. Averyt and H.L. Miller. Cambridge University Press.
- Meinshausen, Malte, Steven J Smith, K Calvin, John S Daniel, M L T Kainuma, J F Lamarque, K Matsumoto, S A Montzka, S C B Raper, K Riahi, and Others. 2011. "The RCP greenhouse gas concentrations and their extensions from 1765 to 2300." *Climatic Change* 109 (1-2):213–241.
- Mercer, John H. 1978. "West Antarctic ice sheet and CO 2 greenhouse effect- A threat of disaster." Nature 271 (5643):321–325.
- Milne, Glenn a., W. Roland Gehrels, Chris W. Hughes, and Mark E. Tamisiea. 2009. "Identifying the causes of sea-level change." *Nature Geoscience* 2 (7):471–478.
- National Research Council. 2013. Abrupt Impacts of Climate Change: Anticipating Surprises.
- National Research Council, Committee on Abrupt Climate Change. 2002. Abrupt Impacts of Climate Change: Anticipating Surprises.
- Nicholls, Robert J. 2011. "Planning for the impacts of sea level rise." Oceanography-Oceanography Society 24 (2):144–157.
- Nicholls, Robert J., P P Wong, V R Burket, J Codignotto, J E Hay, R F McLean, S Ragoonaden, and C D Woodroffe. 2007. "Coastal systems and low-lying areas." .

Nordhaus, William D. 2010a. "DICE-2010 Excel Model."

——. 2010b. "Projections of Sea Level Rise (SLR)."

- Nordhaus, William D. and Joseph Boyer. 2000. Warming the World: Economic Models of Global Warming. MIT Press.
- Nordhaus, William D. and Paul Sztorc. 2013. "DICE 2013R: Introduction and User's Manual." Tech. Rep. April.
- Oppenheimer, Michael. 1998. "Global warming and the stability of the West Antarctic Ice Sheet." *Nature* :325–332.
- Overpeck, Jonathan T. and Julia E. Cole. 2006. "Abrupt Change in Earth's Climate System." Annual Review of Environment and Resources 31 (1):1–31.
- Parson, Edward a. and K Fisher-Vanden. 1997. "Integrated Assessment Models of Global Climate Change." Annual Review of Energy and the Environment 22 (1):589–628.
- Pindyck, Robert S. 2013. "Climate Change Policy: What Do the Models Tell Us?" Journal of Economic Literature 51 (3):860–872.
- Revesz, Richard L., Peter H. Howard, Kenneth Arrow, Lawrence H. Goulder, Robert E. Kopp, Michael A. Livermore, Michael Oppenheimer, and Thomas Sterner. 2014. "Global warming: Improve economic models of climate change." *Nature* 508:173–175.
- Richels, Richard G. and Geoffrey J. Blanford. 2008. "The value of technological advance in decarbonizing the U.S. economy." *Energy Economics* 30 (6):2930–2946.
- Rignot, Eric, G Casassa, P Gogineni, W Krabill, A Rivera, and R Thomas. 2004. "Accelerated ice discharge from the Antarctic Peninsula following the collapse of Larsen B ice shelf." *Geophysical Research Letters* 31 (18).
- Rignot, Eric, J. Mouginot, M. Morlighem, H. Seroussi, and B. Scheuchl. 2014. "Widespread, rapid grounding line retreat of Pine Island, Thwaites, Smith and Kohler glaciers, West Antarctica from 1992 to 2011." *Geophysical Research Letters* 41 (10).
- Rohling, E. J., K. Grant, Ch. Hemleben, M. Siddall, B. a. a. Hoogakker, M. Bolshaw, and M. Kucera. 2008. "High rates of sea-level rise during the last interglacial period." *Nature Geoscience* 1 (1):38–42.
- Rutherford, Thomas F. 2013. "A Minimal Framework with Precautionary Mitigation." In *Climate Change & Integrated Assessment Workshop XIX*. Snowmass, CO.
- Schneider, SH, Kristin Kuntz-Duriseti, and Christian Azar. 2000. "Costing non-linearities, surprises, and irreversible events." Pacific and Asian Journal of Energy 10 (1):81–106.
- Shaffer, G. 2014. "Formulation, calibration and validation of the DAIS model (version 1), a simple Antarctic ice sheet model sensitive to variations of sea level and ocean subsurface temperature." *Geoscientific Model Development* 7 (4):1803–1818.
- Shepherd, Andrew, Duncan Wingham, and Eric Rignot. 2004. "Warm ocean is eroding West Antarctic Ice Sheet." *Geophysical Research Letters* 31 (23).
- Solak, Senay, John-Paul B. Clarke, Ellis L. Johnson, and Earl R. Barnes. 2010. "Optimization of R&D project portfolios under endogenous uncertainty." *European Journal of Operational Research* 207 (1):420–433.
- Stern, Nicholas. 2013. "The structure of economic modeling of the potential impacts of climate change: grafting gross underestimation of risk onto already narrow science models." Journal of Economic Literature 51 (3):838–859.

- Stocker, Thomas F and Andreas Schmittner. 1997. "Influence of CO2 emission rates on the stability of the thermohaline circulation." Nature 388 (6645):862–865.
- Tol, Richard S. J. 2002a. "Estimates of the damage costs of climate change. Part I: Benchmark estimates." *Environmental and Resource Economics* 21:47–73.
 - ——. 2002b. "Estimates of the Damage Costs of Climate Change: Part II. Dynamic Estimates." *Environmental and Resource Economics* 21:135–160.
- Tol, Richard SJ. 1998. "Potential slowdown of the thermohaline circulation and climate policy." Institute for Environmental Studies, Vrije Universiteit Discussion Paper DS98/06.
- Traeger, Christian. 2013. "A 4-stated DICE: quantitatively addressing uncertainty effects in.".
- Urban, Nathan M., Philip B. Holden, Neil R. Edwards, Ryan L. Sriver, and Klaus Keller. 2014. "Historical and future learning about climate sensitivity." *Geophysical Research Letters* 41 (7):2543–2552.
- Urban, Nathan M. and Klaus Keller. 2010. "Probabilistic hindcasts and projections of the coupled climate, carbon cycle and Atlantic meridional overturning circulation system: A Bayesian fusion of century-scale observations with a simple model." *Tellus, Series A: Dynamic Meteorology and Oceanography* 62 (5):737–750.
- van den Broeke, Michiel, Jonathan L Bamber, Janneke Ettema, Eric Rignot, Ernst Schrama, Willem Jan van de Berg, Erik van Meijgaard, Isabella Velicogna, and Bert Wouters. 2009. "Partitioning recent Greenland mass loss." Science (New York, N.Y.) 326 (5955):984–6.
- van Vuuren, Detlef P., Jason Lowe, Elke Stehfest, Laila Gohar, Andries F. Hof, Chris W Hope, Rachel Warren, Malte Meinshausen, and Gian-Kasper Plattner. 2009. "How well do integrated assessment models simulate climate change?" *Climatic Change* 104 (2):255–285.
- Vaughan, David G and John R Spouge. 2002. "Risk estimation of collapse of the West Antarctic Ice Sheet." *Climatic Change* 52 (1-2):65–91.
- Vaughan, DG, JC Comiso, and Ian Allison. 2013. "Observations: cryosphere." Climate
- Webster, Mort, Lisa Jakobovits, and James Norton. 2008. "Learning about climate change and implications for near-term policy." *Climatic Change* 89 (1-2):67–85.
- Webster, Mort, Nidhi Santen, and Panos Parpas. 2012. "An approximate dynamic programming framework for modeling global climate policy under decision-dependent uncertainty." Computational Management Science 9 (3):339–362.
- Weitzman, Martin L. 2009. "On modeling and interpreting the economics of catastrophic climate change." The Review of Economics and Statistics 91 (February):1–19.
- ——. 2012. "The Ramsey Discounting Formula for a Hidden-State Stochastic Growth Process Introduction to Long Term Discounting."
- Weyant, John P., O Davidson, H Dowlabathi, Jae Edmonds, M Grubb, E A Parson, R Richels, J Rotmans, P R Shukla, R S J Tol, and Others. 1996. Integrated assessment of climate change: an overview and comparison of approaches and results. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses: http://www.feem.it/getpage.aspx?id=73&sez=Publications&padre=20&tab=1 http://papers.ssrn.com/sol3/JELJOUR_Results.cfm?form_name=journalbrowse&journal_id=266659 http://ideas.repec.org/s/fem/femwpa.html http://www.econis.eu/LNG=EN/FAM?PPN=505954494

http://ageconsearch.umn.edu/handle/35978

http://www.bepress.com/feem/

NOTE DI LAVORO PUBLISHED IN 2015

ERM	1.2015	Elena Verdolini, Laura Diaz Anadon, Jiaqi Lu and Gregory F. Nemet: <u>The Effects of Expert Selection</u> ,
CCCD	0.0045	Elicitation Design, and R&D Assumptions on Experts Estimates of the Future Costs of Photovoltaics
CCSD	2.2015	James Lennox and Ramiro Parrado: <u>Capital-embodied Technologies in CGE Models</u>
CCSD	3.2015	Claire Gavard and Djamel Kirat: <u>Flexibility in the Market for International Carbon Credits and Price</u> Dynamics Difference with European Allowances
CCSD	4,2015	Claire Gavard: Carbon Price and Wind Power Support in Denmark
CCSD	5 2015	Gunnar Luderer, Christoph Bertram, Katherine Calvin, Enrica De Cian and Elmar Kriegler, Implications of
CCDD	0.2010	Weak Neartern Climate Policies on Longtern Mitigation Pathyays
CCSD	6 2015	Francisca L André and Luis M. do Castro: Incontinues for Drise Manipulation in Emission Dormit Markets with
CCSD	0.2013	Francisco J. Andre and Luis M. de Castro. <u>incentives for Price Manipulation in Emission Permit Markets with</u>
CCCD	7 2015	<u>Stackelberg Competition</u>
CCSD	7.2015	C. Dionisio Perez Blanco and Thomas Thaler: Water Flows in the Economy. An Input-output Framework to
CCCD	0.0015	Assess Water Productivity in the Castile and Leon Region (Spain)
CCSD	8.2015	Carlos M. Gomez and C. Dionisio Pérez-Blanco: Simple Myths and Basic Maths about Greening Irrigation
CCSD	9.2015	Elorri Igos, Benedetto Rugani, Sameer Rege, Enrico Benetto, Laurent Drouet, Dan Zachary and Tom Haas:
		Integrated Environmental Assessment of Future Energy Scenarios Based on Economic Equilibrium Models
ERM	10.2015	Beatriz Martínez and Hipòlit Torró: <u>European Natural Gas Seasonal Effects on Futures Hedging</u>
CCSD	11.2015	Inge van den Bijgaart: <u>The Unilateral Implementation of a Sustainable Growth Path with Directed Technical</u>
		Change
CCSD	12.2015	Emanuele Massetti, Robert Mendelsohn and Shun Chonabayashi: <u>Using Degree Days to Value Farmland</u>
CCSD	13.2015	Stergios Athanassoglou: <u>Revisiting Worst-case DEA for Composite Indicators</u>
CCSD	14.2015	Francesco Silvestri and Stefano Ghinoi : <u>Municipal Waste Selection and Disposal: Evidences from Lombardy</u>
CCSD	15.2015	Loïc Berger: The Impact of Ambiguity Prudence on Insurance and Prevention
CCSD	16.2015	Vladimir Otrachshenko and Francesco Bosello: Identifying the Link Between Coastal Tourism and Marine
		Ecosystems in the Baltic, North Sea, and Mediterranean Countries
FRM	17,2015	Charles E. Mason, Lucia A. Muehlenbachs and Sheila M. Olmstead: The Economics of Shale Gas
2.0.0		Development
FRM	18 2015	Appa Alberini and Charles Towe: Information v. Energy Efficiency Incentives: Evidence from Residential
	10.2015	Alina Albertini and Charles Towe. Internation V. Energy Enclency Incentives. Evidence from Residential
CCSD	10 2015	Zhong Viang Zhang Charles the Diversity for the Stanger The Case of Cathon Trading in China
CCSD	19.2013	2100gAlang Zhang, <u>Crossing the Kiver by reening the Stones</u> , the <u>Case of Carbon trading in China</u>
CCSD	20.2015	Petterson Wolma Vale: <u>The Conservation Persus Production Trade-off: Does Livestock Intensincation</u>
CCCD	21 2015	Increase Detorestation? The Case of the Brazilian Amazon
CCSD	21.2015	Valentina Bosetti, Melanie Heugues and Alessandro Tavoni: <u>Luring Others into Climate Action: Coalition</u>
CCCD		Formation Games with Threshold and Spillover Effects
CCSD	22.2015	Francesco Bosello, Elisa Delpiazzo, and Fabio Eboli: <u>Macro-economic Impact Assessment of Future Changes</u>
		in European Marine Ecosystem Services
CCSD	23.2015	Maryse Labriet, Laurent Drouet, Marc Vielle, Richard Loulou, Amit Kanudia and Alain Haurie: Assessment of
		<u>the Effectiveness of Global Climate Policies Using Coupled Bottom-up and Top-down Models</u>
CCSD	24.2015	Wei Jin and ZhongXiang Zhang: <u>On the Mechanism of International Technology Diffusion for Energy</u>
		<u>Technological Progress</u>
CCSD	25.2015	Benjamin Michallet, Giuseppe Lucio Gaeta and François Facchini: <u>Greening Up or Not? The Determinants</u>
		Political Parties' Environmental Concern: An Empirical Analysis Based on European Data (1970-2008)
CCSD	26.2015	Daniel Bodansky, Seth Hoedl, Gilbert Metcalf and Robert Stavins: Facilitating Linkage of Heterogeneous
		Regional, National, and Sub-National Climate Policies Through a Future International Agreement
CCSD	27.2015	Giannis Vardas and Anastasios Xepapadeas: Time Scale Externalities and the Management of Renewable
		Resources
CCSD	28 2015	Todd D. Gerarden Richard G. Newell, Robert N. Stavins and Robert C. Stowe: An Assessment of the
CCDD	20.2010	Energy-Efficiency (an and its implications for Climate Change Policy
CCSD	20 2015	Criticity Circuity and Emanuela Massetti Micration and Climate Change in Pural Africa
EDM	29.2015	Cinsula Catalico alla Emandele Massetti. Magration and Cimitate Change in Tura Antica.
ERIVI	30.2013	Simone ragiapierra. The Future of Kenewable Energy in the Mediterranean. Translating Potential into
CCCD	21 2015	
CCSD	31.2015	Jan Siegmeier, Linus Mattauch, Max Franks, David Kienert, Anselm Schultes and Ottmar Edenhofer: <u>A Public</u>
CCCP	20.001-	Finance Perspective on Climate Policy: Six Interactions That May Enhance Welfare
CCSD	32.2015	Rever Gerlagh, Inge van den Bijgaart, Hans Nijland and Thomas Michielsen: <u>Fiscal Policy and CO2 Emissions</u>
		ot New Passenger Cars in the EU
CCSD	33.2015	Marie-Laure Nauleau, Louis-Gaëtan Giraudet and Philippe Quirion: Energy Efficiency Policy with Price-
		quality Discrimination

CCSD	34.2015	Eftichios S. Sartzetakis, Anastasios Xepapadeas and Athanasios Yannacopoulos: <u>Regulating the</u>
		Environmental Consequences of Preferences for Social Status within an Evolutionary Framework
CCSD	35.2015	Todd D. Gerarden, Richard G. Newell and Robert N. Stavins: <u>Assessing the Energy-efficiency Gap</u>
CCSD	36.2015	Lorenza Campagnolo and Fabio Eboli: <u>Implications of the 2030 EU Resource Efficiency Target on</u>
		Sustainable Development
CCSD	37.2015	Max Franks, Ottmar Edenhofer and Kai Lessmann: Why Finance Ministers Favor Carbon Taxes, Even if They
		Do not Take Climate Change into Account
CCSD	38.2015	ZhongXiang Zhang: Carbon Emissions Trading in China: The Evolution from Pilots to a Nationwide Scheme
CCSD	39,2015	David García-León: Weather and Income: Lessons from the Main European Regions
CCSD	40 2015	Jaroslav Mysiak and C. D. Pérez-Blanco: Partnerships for Affordable and Equitable Disaster Insurance
CCSD	41 2015	Surminski ICIH Aerts WIW Botzen P Hudson I Mysiak and C D Pérez-Blanco: Reflections on the
CCDD	11.2010	Current Debate on How to Link Flood Insurance and Disaster Risk Reduction in the European Union
CCSD	42 2015	Erin Baker, Olaitan Olaleve and Lara Aleluia Reis: Decision Frameworks and the Investment in R&D
CCSD	42.2015	C. D. Storz Blacco and C. M. Cómer: Development the Willington provide the Investment in Agriculture
CCSD	43.2015	C. D. Felezblanco and C. M. Gomez. <u>Revealing the Winning less to Fay for income instruction Agriculture</u>
CCSD	44.2013	Banchongsan Charbensook. On the Interaction between Player neterogeneity and Partner Neterogeneity in
CCCD	45 2015	I wo-way Flow Strict Mash Networks
CCSD	45.2015	Erin Baker, Valentina Bosetti, Laura Diaz Anadon, Max Henrion and Lara Alelula Reis: <u>Future Costs of Key</u>
		Low-Carbon Energy Technologies: Harmonization and Aggregation of Energy Technology Expert Elicitation
		<u>Data</u>
CCSD	46.2015	Sushanta Kumar Mahapatra and Keshab Chandra Ratha: <u>Sovereign States and Surging Water: Brahmaputra</u>
		<u>River between China and India</u>
CCSD	47.2015	Thomas Longden: CO ₂ Intensity and the Importance of Country Level Differences: An Analysis of the
		Relationship Between per Capita Emissions and Population Density
CCSD	48.2015	Jussi Lintunen and Olli-Pekka Kuusela: <u>Optimal Management of Markets for Bankable Emission Permits</u>
CCSD	49.2015	Johannes Emmerling: <u>Uncertainty and Natural Resources - Prudence Facing Doomsday</u>
ERM	50.2015	Manfred Hafner and Simone Tagliapietra: <u>Turkish Stream: What Strategy for Europe?</u>
ERM	51.2015	Thomas Sattich, Inga Ydersbond and Daniel Scholten: <u>Can EU's Decarbonisation Agenda Break the State-</u>
		Company Axis in the Power Sector?
ERM	52.2015	Alessandro Cologni, Elisa Scarpa and Francesco Giuseppe Sitzia: <u>Big Fish: Oil Markets and Speculation</u>
CCSD	53.2015	Joosung Lee: Multilateral Bargaining in Networks: On the Prevalence of Inefficiencies
CCSD	54.2015	P. Jean-Jacques Herings: <u>Equilibrium and Matching under Price Controls</u>
CCSD	55.2015	Nicole Tabasso: Diffusion of Multiple Information: On Information Resilience and the Power of Segregation
CCSD	56.2015	Diego Cerdeiro, Marcin Dziubinski and Sanjeev Goyal: Contagion Risk and Network Design
CCSD	57.2015	Yann Rébillé and Lionel Richefort: <u>Networks of Many Public Goods with Non-Linear Best Replies</u>
CCSD	58.2015	Achim Hagen and Klaus Eisenack: International Environmental Agreements with Asymmetric Countries:
		Climate Clubs vs. Global Cooperation
CCSD	59.2015	Ana Mauleon, Nils Roehl and Vincent Vannetelbosch: Constitutions and Social Networks
CCSD	60.2015	Adam N. Walker, Hans-Peter Weikard and Andries Richter: The Rise and Fall of the Great Fish Pact under
		Endogenous Risk of Stock Collapse
CCSD	61,2015	Eablo Grazi and Henri Waisman: Agglomeration, Urban Growth and Infrastructure in Global Climate Policy:
0000	0112010	A Dynamic CGE Approach
CCSD	62 2015	Flori loss Repedetto Rugani Sameer Rege Enrico Benetto Laurent Drouet and Dan Zachary: Combination
CCSD	02.2013	of Equilibrium Models and Hybrid Life Cycle-Input-Output Analysis to Predict the Environmental Impacts of
		Energy Doling Coopering
	63 2015	Delayane B. Diaz: Estimating Clobal Damages from Sea Lovel Pice with the Coastal Impact and Adaptation
CCSD	03.2013	Model (CIAM)
CCSD	64 2015	Noder (Civing) Delayang B. Diaz Integrated Accordment of Climate Catactrophes with Endogeneus Uncertainty Deep the
CCDD	04.2013	Delavane D. Diaz. Integrated Assessment of Chinate Catastrophes with Endogenous Oncertainty: Does the
		NISK OF ICE SHEEL CONADSE JUSTIV FREGAUTIONARY MILLIVATION?